



U.S. Department of Labor U.S. Bureau of Labor Statistics Office of Productivity and Technology

R&D and Productivity Growth: A Review of the Literature

Leo Sveikauskas, U.S. Bureau of Labor Statistics

Working Paper 408 September 2007

All views expressed in this paper are those of the author and do not necessarily reflect the views or policies of the U.S. Bureau of Labor Statistics.

R&D and Productivity Growth: A Review of the Literature¹

Leo Sveikauskas Division of Productivity Research and Program Development Bureau of Labor Statistics, Washington, D.C. 20212

Bureau of Labor Statistics Working Paper 408

Abstract

This paper reviews the literature on R&D to provide guidelines for recent efforts to include R&D in the national income accounts. The main conclusions are:

1. Measures of R&D as an asset held by a particular owner must be complemented by estimates of the spillover effect of R&D in order to obtain a reliable measure of the overall effect of R&D on productivity growth.

2. If research financed by the government and research financed by business are both counted as investment, some double counting occurs and growth accounting analysis overstates the role of research relative to other factors.

3. The overall rate of return to R&D is very large, perhaps 25 percent as a private return and a total of 65 percent for social returns. However, these returns apply only to privately financed R&D in industry. Returns to many forms of publicly financed R&D are near zero.

4. Firm R&D should be allocated to the different industries in which a firm produces, rather than all credited to the firm's main industry. An allocation procedure is proposed.

5. Much further work needs to be carried out to understand how R&D conducted in the richest countries is transmitted to developing countries. Detailed microeconomic data on firms or establishments in developing nations will be necessary to understand the channels of technology transfer more fully.

Journal of Economic Literature categories: O30, O40.

Keywords: R&D stocks, R&D spillovers, R&D and productivity growth

¹ This paper reviews several portions of the scholarly literature on research and development which are relevant to current efforts to establish R&D satellite accounts. I appreciate comments from Michael Harper, Peter Meyer, Carol Robbins, and Larry Rosenblum, and remarks from seminar participants at the Bureau of Economic Analysis and the Bureau of Labor Statistics. However, the views expressed in this paper are solely those of the author, and do not reflect the views or policies of the U.S. Bureau of Labor Statistics or the Bureau of Economic Analysis.

R&D and Productivity Growth: A Review of the Literature

Leo Sveikauskas² Bureau of Labor Statistics

Recent years have seen substantial progress towards including research and development (R&D) as a capital investment within the national income accounts (Canberra Group II (2003)). Economists at the United States Bureau of Economic Analysis, or BEA, (Carson, Grimm and Moylan (1994), Fraumeni and Okubo (2005); Okubo <u>et al</u> (2006b)) have prepared initial versions of a potential R&D satellite account. The U.S. National Science Foundation has agreed to support BEA's further work on R&D accounts, and the BEA is committed to publish R&D accounts for the United States. Economists in Australia (Australian Bureau of Statistics (2004)) and in the Netherlands (de Haan (2004)) have also reported initial R&D stocks for their countries.

One main motive for adding R&D is to broaden the accounts to include a further important source of economic growth. Present international discussion has largely concentrated on treating R&D as an asset for private firms, the government, universities, and other nonprofit organizations. However, most gains from R&D typically do not accrue to the original investors. Benefits instead spill over to a wide variety of other economic agents. Consequently, it is necessary to establish links between the role of R&D as an asset and the total effect of R&D on economic growth, which reflects both asset returns and spillovers.

² The views expressed here are those of the author, and do not represent the Bureau of Labor Statistics.

Many central characteristics of R&D stocks, such as the rate of depreciation, the lag until R&D becomes operative, and the rate of return, differ widely between the asset and spillover effects. In a sense, analysts have to prepare two different sets of calculations to describe the asset and spillover effects of R&D. From this perspective, the stocks of R&D assets the U.S. Bureau of Economic Analysis is currently preparing neatly complement the spillover R&D stocks which the U.S. Bureau of Labor Statistics has published for many years.³ Other countries currently measuring R&D assets will similarly have to expand their analyses to include spillovers in order to describe the full impact of R&D on productivity growth.

This paper examines the relationship between R&D as an asset and R&D as an ingredient in growth. Section I considers recent measures of asset and spillover R&D stocks in the United States, and also examines which portions of R&D should be included in national R&D stocks. Section II reviews the economics literature on the private and social (private plus spillover) impact of R&D. Section III considers some difficulties associated with measuring R&D accurately in each industry. Section IV examines the international transmission of R&D, a major source of spillovers for most of the countries in the world. Section V concludes.⁴

I. <u>Introduction: Understanding Asset and Spillover Returns to R&D; What Portions of R&D Should be Included in National R&D Stocks?</u>

3

³ Okubo <u>et al</u> (2006b) describe BEA work on R&D as an asset, and U.S. Bureau of Labor Statistics (1989) presents the BLS work on R&D spillovers.

⁴ An earlier version of the present paper was prepared as a background paper for the Canberra II discussion of research and development.

Most Canberra II group discussion of R&D concentrated on similarities between R&D and other assets already recognized as capital.⁵ For example, firms commit resources to R&D in the hope that this investment will create an asset which eventually lead to greater profits, much as in the case of plant or equipment.

The comparison is valid, and supports the proposal that R&D expenditures should similarly be treated as capital investment. However, R&D differs from other forms of capital investment in several respects. A central characteristic is that payoffs from R&D are not limited to the original investors, but also accrue to competitors, other firms, suppliers, and customers.⁶ Consequently, it is too narrow to view R&D as essentially an asset held by a particular owner. A broader perspective on R&D has to include knowledge spillovers, which have often been a central theme in discussions of productivity growth.⁷

To illustrate the importance of the diffusion of knowledge from firms undertaking research to the broader community, consider technical change in pharmaceuticals. Statins are a new class of anti-cholesterol drugs which have contributed greatly to the decline of heart disease. A major pharmaceutical firm introduced the first commercial

⁵ The March 2004 meeting of the Canberra II group, held in Washington, discussed R&D as an asset, and concentrated on similarities between R&D and other assets.

⁶ Romer (1990, pp. 73-76) discusses nonrival goods, for which the use by one firm or person does not limit its use by another. Romer also considers excludability, and points out that the results from research can only partially be excluded from use by others.

Section II reviews the literature on the effect of R&D on productivity growth.

In addition to the nonrival characteristic of R&D knowledge, it is typically difficult to determine the value of output from research because research results are typically not sold on the open market. Economists have measured research outcomes from statistical relationships between R&D and several indicators of technological success. Most studies determine the return to research from its relationship with productivity growth. Other work evaluates the effectiveness of research from its connection with profits or stock market value.

statin product in 1987, and conducted pioneering research demonstrating that statins were safe, lowered cholesterol, and successfully reduced the death rate from heart disease. Since 1987, several firms have introduced new and improved statins. A different firm now produces a new and greatly improved statin, which lowers cholesterol more effectively, and has therefore become the market leader.

Although the second firm now dominates the market for statins, it is <u>not</u> the case that the second firm's private investment in R&D is now the only relevant R&D. From the point of view of private returns, much of the early research which the initial firm carried out is indeed no longer profitable.⁸ However, in a broader sense all the initial research which demonstrated that statins were safe, highly effective, and reduced the incidence of heart disease still provides the core knowledge of the present day industry. The first firm's initial investment in R&D is still relevant to the industry and still provides important social returns, even though most of the private returns now go to the second firm.

To take another example, two leading firms have competed in producing microprocessors for many years. When the technology leader introduces a new chip, the second firm soon matches, and prices fall rapidly. As a result, microprocessor prices have declined sharply. Most of the benefits of innovation have been captured by consumers through lower prices. The profit of the innovators, obtained through returns to the R&D they conduct, is only a small part of the picture.

5

⁸ In fact, the initial drug, introduced in 1987, is no longer under patent.

These examples illustrate how the knowledge and benefits obtained from R&D typically leak out from the original performers of R&D to competitors, to other firms, to consumers, and, eventually, to other countries. Many forms of knowledge are useful to other firms (and so have a social return) even when they no longer pay off to the firm initiating the research (no longer have a private return). Similarly, consumers obtain better or cheaper products (benefit from social returns) even if the private return to firms turns out to be low.⁹

As knowledge gradually leaks out, private benefits decline and spillover effects increase. Consequently, private and spillover returns follow different time paths. Quite reasonably, spillover effects are considerably more long lived than private effects. Okubo <u>et al</u>. (2006b) calculate R&D asset stocks assuming a 15 percent (or greater) annual depreciation rate. In contrast, the Bureau of Labor Statistics (1989), measuring the longer lasting spillover effects, assumes 10 percent depreciation for applied research and development and zero depreciation for basic research, which implies an overall depreciation rate of less than 9 percent.¹⁰

Such substantial differences in depreciation have an important impact on the implied size of R&D stocks and the time pattern of anticipated benefits. For example, in 2000 the BLS spillover R&D stock is 44 percent greater than it would have been if calculated at 15 percent depreciation. A 15 percent depreciation rate implies that a quarter of an

6

⁹ Measured benefits are limited to those which have a market evaluation. Some benefits, such as clean air or some types of medical advances, are not evaluated through market prices, and are typically not included in economic statistics.

¹⁰ Since the BLS assumes zero depreciation of basic research, the basic research proportion of the overall BLS stocks is greater than data on annual R&D investments would suggest.

investment has depreciated after two years, half by five years, and three-quarters by 9 years. In contrast, 9 percent depreciation indicates a quarter of investment has depreciated by 4 years, half by 8 years, and three-quarters by 15 years.

The influence of depreciation is accentuated by differences in the lags with which R&D investment enters the R&D stock. BEA R&D investments enter the asset stock immediately, whereas BLS investments enter the spillover stock with a lag of two years for applied research, and five years for basic research.¹¹ The longer lags further imply that spillovers occur later than asset returns.

There are also major differences in the returns to asset and spillover R&D. Somewhat surprisingly, it is quite difficult to determine exactly how much of R&D returns accrue through private returns and how much through spillovers. Most of the economics literature typically reports the amount by which the average social return to R&D exceeds the private return.¹² Remarkably few studies (Schankerman (1981); Cuneo and Mairesse (1984)) actually examine the actual return to private firms, which is most relevant in considering R&D as an asset.

¹¹ In the BEA stocks, if investment occurs in 1990, half enters the R&D stock at the start of 1990 and half at the start of 1991. In the BLS calculations, 1990 investment in applied research enters the research stock at the start of 1992.

¹² Studies typically include the capital, labor, and other factors used in research within the inputs used to measure productivity growth. Therefore, the private value of research is removed from such productivity measures. Consequently, when R&D is used to explain productivity growth, the estimated return is the amount by which the return to R&D exceeds the private value of these resources. Griliches and Lichtenberg (1984b, page 475) and U.S. Bureau of Labor Statistics (1989, page 36) provide a more formal statement of this argument.

In other relevant literature, Nordhaus (2004) emphasizes that only a very small portion of the returns to R&D accrues as abnormally high returns to the innovator. Baumol (2000) describes how firms continually improve the quality of their products, through R&D, and leapfrog beyond the products produced by their competitors' R&D; this churning process reduces the value of their competitors' private R&D.

Nevertheless, the existing literature generally finds that spillovers are greater than private returns. Griliches (2000, page page 70) concludes that the excess return to firms¹³ is 10 percent, whereas the social return, which includes private returns, is 25 percent. Section II of the present paper reviews the evidence and suggests the private return to R&D is 25 percent, while the social return is 65 percent. Okubo <u>et al.</u> (2006a, page 27) examine many different studies, and conclude that the private return is 26 percent and the social return 66 percent. Each of these three judgements indicates spillovers contribute approximately three-fifths of the total return to R&D.

Of course, the conclusion that spillover effects are primary is necessarily tentative because the available evidence is incomplete. Clearly, most studies do not include the private value of resources used in research, and thereby understate the contribution of R&D.¹⁴ Studies of specific innovations offer an alternative method of measuring actual returns to private firms; subsection II.5 below examines evidence from the innovation studies and reaches roughly similar conclusions concerning the relative importance of private and spillover effects. In addition, most studies omit the international effects of R&D (Coe and Helpman (1995)) and thereby miss an important element of spillovers.

Another central question in the capitalization of R&D is which portions of R&D should be included as investment. The current consensus is that all private, government,

¹³ Footnote 10 explains why the R&D literature typically reports excess returns which exceed the private return.

¹⁴ Since most studies report only a portion of total returns to R&D (the excess return), the implied total effect of R&D will clearly be greater if the value of the private inputs used in R&D is also included.

In this context, a BEA analysis of industry data (Schultz (2006)) has the great virtue that it isolates the capital, labor, and materials used in R&D and removes these inputs from the factors used in production. If similar adjustments could be made within firm data, it would be possible to obtain better estimates of actual overall returns to each firm's private R&D.

or university R&D should be treated as investment.¹⁵ However, two issues arise here. First, many forms of R&D have little economic value and do not contribute to growth. Should they all be counted as investment? Second, if publicly financed basic research and subsequent private research are both treated as investment, a certain amount of double counting occurs. The rest of Section I examines these two topics.

In considering which forms of R&D should be designated as investment, the asset and growth views of R&D once again offer contrasting perspectives. The 1993 System of National Accounts defines an asset, in part, by requiring that "Economic benefits may be derived by their owners by holding them or using them over a period of time". In this definition, the word "may" leads to an extremely broad definition of assets. Even if a broad category of research spending brings almost no demonstrated economic returns, just the possibility that a very small portion of these expenditures "may" be able to bring returns leads to the entire sum being classified as an asset. So it is not surprising that current discussion suggests that all R&D expenditures by firms, universities, and governments (as R&D is defined in the OECD Frascati Manual) should be treated as investment.

In contrast, from the point of view of growth, what matters is the proportion of an expenditure category which can be expected to provide returns and the magnitude of the returns. From this perspective, most discussion has emphasized privately financed research, which has been amply demonstrated to bring high returns. Of course, as Scherer (1999) pointed out, most private R&D projects do not actually earn positive

9

¹⁵ Canberra II group (2006); Robbins (2006); Okubo <u>et al</u> (2006b).

returns. Nevertheless, all private R&D expenditures can plausibly be treated as investment because these expenditures are all undertaken with the expectation of a commercial return and "may" obtain such returns.

In contrast, broad slices of university and government research are conducted with very little expectation or prospects of direct commercial returns.¹⁶ For example, only a very small part of the research conducted by universities and colleges brings commercial returns.¹⁷ As President Richard Levin of Yale University recently remarked "We're not trying to drive university science by commercial objectives. We want to do great science. Some of that will have commercial potential; most of it won't." Also "Our technology transfer strategy is not to maximize revenue. Our primary goal is to get the findings of our laboratories out into practice."¹⁸

Federally financed industrial research accounted for more than half of industrial research in the late 1960s, though this proportion is now less than one tenth. Evidence on the productivity impact of federal research in industry is mixed. The industry studies and the best study of firm data (Leonard (1971); Terleckyj (1974); Lichtenberg and Siegel

¹⁶ Fraumeni and Okubo (2005) assume the return to public research is two-thirds of the return to private research, which is clearly an overstatement because they are working within the framework of the current national income accounts, which do not include the returns to federal investment in R&D within output.

More recently, government statisticians have decided to include returns to government capital as output, and it is clearly appropriate to include returns to government R&D stocks as well. Nevertheless, as the text points out, many forms of federal and university research have little economic value and do not contribute to growth. Even though such expenditures are classified as R&D, I would argue that, because there is no evidence of a positive return, such expenditures are actually a form of government consumption, often for quite useful purposes, but do not deserve to be treated as investment.

¹⁷ Even universities as eminent as Michigan, Rochester, and Yale earn surprisingly small amounts from patent royalties. For example, over the last decade Yale earned approximately \$21 million dollars a year from patent royalties

¹⁸ <u>Yale Alumni Bulletin</u>, November 2006.

(1991)) conclude federal research in industry does not improve industrial productivity.¹⁹ In contrast, Griliches (1986) concluded federal R&D in industry had a positive effect on productivity, though less of an impact than privately financed research. Even beyond university and federally financed industrial research, Piekarz (1983) reviews evidence indicating that many federally funded research programs have not been very effective.

It is disturbing to assign investment status to broad categories of research spending, such as university research or federally financed R&D in industry, when the fundamental evidence indicates that, though returns "may" exist, actual returns are unlikely or at best accrue to only a small proportion of expenditures. In my judgement, large batches of expenditures where direct economic value is unlikely should not be treated as R&D investment, on the same basis as private R&D, merely on the off chance that some small amounts of these expenditures "may" turn out to have economic value.²⁰ If all expenditures in government and university R&D are treated as investment, the true

¹⁹ In the United States, federally financed research expenditures refer to research financed by the national, or federal, government. On the basis of the type of evidence summarized in the text, the U.S. Bureau of Labor Statistics (1989), in its work on productivity growth, chose to include only privately financed R&D conducted in industry in its measures of the relevant R&D stock.

Lichtenberg and Siegel (1991) is the best study of these matters within firm data because their paper adjusts the firm output deflators to reflect the different industries in which each firm actually produces.

Some papers find that federally financed research in industry has a positive effect. Bonte (2003) argues that federally financed research contributes to productivity growth in the United States, but his analysis is based on a cointegration analysis of aggregate data, and does not use the highly detailed firm data analyzed in Lichtenberg and Siegel (1991).

A potentially important recent study (Guellec and van Pottelsberge de la Potterie (2004)) argued that government research expenditures, given private R&D, contribute to the rate of economic growth. This analysis is based on aggregate data for 16 OECD countries. Such conclusions run counter to most of the literature, which indicates that public research affects measured growth only through its effect on private research (OECD (2003) in addition to the U.S. studies mentioned in the text). Without more detailed microeconomic evidence establishing specific channels through which public science affects productivity growth in these 16 countries (such as a more rapid growth of high-tech industries), the direct importance of government R&D in growth remains an open question.

²⁰ Consider a similar argument in the case of an auto dealer who is selling 100 cars, of which one or two "may" be used as an investment. In this context, it does not make much sense to consider all 100 cars as investment. Yet this is fundamentally what is being proposed in designating all university and government research as investment.

amount of relevant R&D investment is greatly overstated.²¹ In my judgement, any procedure which counts all government or university research as equivalent, on a dollar for dollar basis, to private research spending provides an extremely flawed measure of how much economically relevant research investment actually occurs.

Of course, some government and university research has been greatly successful. Several programs, especially those in which university scientists compete for grants, such as the National Science Foundation, the National Institues of Health, some Department of Agriculture programs, and DARPA in the Department of Defense, appear to have a remarkable record. In addition, much federal and university research is conducted on health care, and Nordhaus (2003) demonstrated the importance of such work. Similarly, much military research substantially improved military effectiveness, as any comparison of the accuracy of precision guided missiles in the Gulf War of the early 1990s and in the Iraq war of the early 2000s clearly indicates.

However, these government successes have been influential primarily through their impact outside the public sector. For example, when physicists developed better methods of transmitting their papers through ARPANET, this precursor to the Internet did not greatly affect government output. Except for some elements of military research, each of

As an illustration of the importance of properly defining the boundaries of research investment, in Lisbon the European Union adopted a goal that R&D expenditures should be three percent of GDP. Private R&D expenditures are falling far short of this target. The Europeans may therefore consider greater "investment" in university or government R&D. I would argue that the economic returns to public R&D are far lower than those to private R&D and that, if the Europeans follow such a strategy, they will be badly disappointed with the returns. Overall returns for university and government research are low primarily because so much research in these fields has no direct financial payoff.

these major successes has probably had its greatest economic impact through the private sector.

The discussion above demonstrated much university and federal research has little commercial value. On the other hand, perhaps federal R&D produces more government output, and should therefore perhaps be assigned a rate of return, such as the Treasury bill rate. Such suggestions are inherently plausibile. Perhaps R&D is effective in a variety of circumstances, and brings government agencies useful benefits, even if these are currently not included in GDP. However, proposals such as these determine which R&D is useful by assumption rather than through evidence. If the crucial contribution of federal R&D to growth is through increasing government output, advocates of including federally financed R&D, and its returns, in the accounts need to demonstrate that federal R&D actually brings about increased federal output.²² Lehr and Lichtenberg (1998) showed it is possible to determine how new technology improves the productivity of government agencies. If this channel is crucial in justifying federal R&D as an investment, we need much better evidence on how federally financed R&D improves federal productivity, and on whether there are important areas in which federal R&D is ineffective, before federally financed R&D, and its assumed returns, can be included in the national accounts.

13

²² As Section II below illustrates, the notion that private R&D contributes to growth is supported by a rich variety of evidence which greatly clarifies the returns to private R&D. If evidence did not consistently show that private R&D affects productivity and output growth, the international statistical community would probably not now be supporting the inclusion of private R&D as an investment. Surely, it is just as necessary to have sound empirical evidence demonstrating that there are returns to federal R&D before such expenditures are included as an asset.

It is not sufficient to finesse these points by simply assigning a lower rate of return to all university and federal research. The more fundamental problem is that many forms of research, however worthy they may be, do not deserve to be designated as investment because they do not contribute to the future growth of output.

Of course, many advances arising from university or government research eventually have an important indirect effect on growth. Consider, however, the usual conventions and methods of growth accounting. Growth accounting typically examines only the proximate sources of growth. For example, growth due to a higher number of educated workers is typically attributed to labor quality, and growth due to larger amounts of capital is similarly attributed to capital formation. Even if improvements in technology are the fundamental underlying cause, which creates the demand for a more educated work force or creates the new equipment which makes further capital deepening feasible, growth accounting still attributes these observed output gains to education and to capital deepening rather than to technology. In a similar way, the flow of understanding from federal or university research typically operates by increasing the demand for private R&D. If the National Institutes of Health develop new insights into the treatment of a particular disease, large amounts of further private R&D are usually required to convert these new ideas into new pharmaceuticals (Toole (2007)). Consistent with the treatment of other inputs in growth accounting, growth accounting should include only the private R&D and not the underlying public or university research which makes the private effort feasible. If we count both the underlying government and university research and also

the proximate private R&D spending, we are in a sense double counting the investment in R&D in comparison with our treatment of other inputs.²³

There may be some argument for double counting in the case of military technology, where improved weapons directly meet an important government goal, and the spillovers involved in dual-use technology are peripheral. However, for other government research, in such areas as health, science, technology, and agriculture, the fundamental government goals are downstream, such as the introduction of better medicines or the promotion of industrial growth and technology.²⁴ In the context of growth accounting, it is not appropriate to include here both the public research which establishes understanding and the private research which implements the new concepts.

In summary, if we are operating within the framework of growth accounting, only a relatively small proportion, perhaps a quarter or less, of the R&D conducted in government or in universities can be expected to have commercial returns or to bring firmly established imputed returns. Such a proportion is sharply lower than the two-thirds suggested by Fraumeni and Okubo (2005).²⁵

²³ Recent empirical studies have established that certain broad economic features, such as legal protections for private property (Scully (1988), La Porta, Lopez-de-Silanes, Shleifer, and Vishny (1998), Djankov, La Porta, Lopez-de-Silanes, and Shleifer (2003)) or the presence of a large financial sector (King and Levine (1993)), have a positive impact on growth. In much the same way, high quality government or university research is likely to create a favorable framework for growth. Nevertheless, economists typically do not treat government expenditures on the legal or financial system as an investment.

In contrast, government investment in transportation infrastructure, where benefits can plausibly be assigned to individual firms, is often included in productivity studies.

²⁴ Some may argue that the government is really trying to advance science and knowledge for its own sake, which is a useful goal. However, if the process did not eventually lead to better treatment for diseases and a longer life span, funding for NIH would be much lower.

²⁵ As some readers have suggested, by this logic if a research team in a medical school is hired to do similar work in a pharmaceutical firm, such a change in the work environment will increase measured

II. <u>The Literature on the Rate of Return to R&D</u>.

The effect of R&D on the economy depends on both what expenditures are included in R&D, as discussed in Section I, and what the returns to R&D are. Section II therefore reviews evidence on the returns to R&D. The rate of return to R&D can be examined at many different levels of aggregation, such as the returns to individual research projects, the returns to firms, returns to an industry, returns to a national economy, or even returns within the entire world.

II.1 Private (Firm Level) Returns to R&D.

The return to research has generally been estimated (Griliches (1980)) by comparing productivity growth or profitability in different firms with research expenditures or the growth of the research stock within these firms.²⁶ Nadiri (1993, Table 1a) reports the results from many studies which examined the return to R&D within data for individual firms.

Several difficulties limit the validity of estimates of the private return to R&D. Substantial complementary resources are often used with R&D. For example, the results from new research have to be put into effective operation. In many cases, producing a new product requires a pilot plant or other new methods of production. This requires

national investment. The counterargument is that the firm presumably selects this research team, out of many available at the med school, because their work is atypical and has the greatest commercial potential. ²⁶ Lichtenberg and Siegel (1991) report estimates of the return to R&D based on productivity data for U.S. firms. The Lichtenberg-Siegel estimates are based upon a large sample of firms and greatly improved estimates of firm productivity. Griliches (2000, pp. 65-66) summarizes some of the main results concerning the stock market's evaluation of firm's R&D efforts. Hall, Jaffe, and Trajtenberg (2005) show that highly cited patents strongly affect stock market evaluation, but less cited patents have little impact.

substantial engineering work. In addition to the capital costs involved, managers and workers will have to be trained in the new techniques. Similarly, new products typically require very close coordination with marketing specialists. The potential market has to be evaluated, and salesmen have to be hired and trained to sell the new product.²⁷ The high returns firms apparently earn on research expenditures may partially reflect returns to such complementary investments.

In addition, returns to private firms may be high because they include a substantial risk premium. Mansfield <u>et al</u>. (1971) and Scherer (1999) show R&D projects have considerable technical and commercial risk and many projects are not very successful. Observed returns to R&D could, in part, reflect a substantial risk premium.

Nelson (1988) raised several important issues which should make analysts quite careful about accepting and using typical estimates of the rate of return to R&D. Nelson points out that R&D and technological opportunity are closely intertwined. Firms with high R&D typically have more favorable technological opportunities. Similarly, firms with low R&D are likely to face weaker technological opportunities. Consequently, it is difficult to determine the extent to which effects attributed to R&D expenditures actually reflect the collateral technological opportunities. For this reason, Nelson argues that standard regressions, across firms or industries, cannot provide plausible estimates of the rate of return to research investments.^{28,29}

²⁷ Mansfield <u>et al</u>. (1971, pp. 112-114) describes the many different types of additional inputs which are also necessary in the process of innovation.

²⁸ Griliches (1979, Section 5) also warned about simultaneity problems arising from the mutual dependency between R&D investments and past and future expected output. Olley and Pakes (1996)

II.2 Industry (Direct) Returns to R&D.

Studies of the direct return to research examine the relationship between R&D and productivity within data for different industries. By examining industry data, analysts are able to include spillovers between the different firms within an industry.

There are important differences between firm and industry returns to R&D.³⁰ As

Griliches (1979, page 101) remarks, firm knowledge leaks out fairly rapidly, so firm

R&D depreciates rapidly. On the other hand, much of this knowledge is gained by other

firms, so social returns depreciate more slowly than firm returns. Similarly, there is a

greater lag before social returns take effect, and social returns are substantially greater

than private returns. Consequently, many of the central features of R&D stocks are

Fung (2004) measures intraindustry and intrafirm spillovers from patent citations, and includes measures of the overlap in patent citations and the scope (number of patent classes involved in a firm' research) of patents. He finds several of these influences are significant in explaining the effectiveness of R&D. Furthermore, allowing for technological opportunity generally increases estimates of the R&D elasticity. Fung also considers some of the relationships between his data and the technological characteristics of industries. Any progress in this direction is very welcome, but a more fundamental understanding of how industry characteristics affect R&D patterns in a broad group of industries is necessary before we can understand why technological opportunities differ so widely among industries. Klevorick, Levin, Nelson, and Winter (1995) take an interesting initial step in this direction.

³⁰ Many R&D deflators (U.S. Bureau of Labor Statistics (1989); Okubo <u>et al.</u> (2006b)) combine labor costs and output price indices, which permits R&D deflators to rise less rapidly than input costs, reflecting increases in productivity. However, Griliches (1990) and especially Kortum (1997) show that over time greater amounts of research are necessary to generate a new patent. Though there are valid questions concerning the use of patents as a measure of research output and how the economic meaning of a patent has changed over time, this evidence suggests that the productivity of R&D does not increase over time.

suggest an important method of addressing such concerns about simultaneity, but do not apply their procedure to R&D. Acharya and Keller (2007) usefully apply some of the Olley-Pakes ideas to R&D. Such methods are important because they provide estimates of the return to R&D which are much less subject to the problem of endogeneity.

²⁹ If R&D is thought to reflect some other influence, the standard econometric procedure would be to attempt to find an instrumental variable which influences R&D growth but is independent of the productivity of R&D. Barro (1998, p. 20) comments "R&D spending may respond to exogenous changes in productivity growth so that the estimated coefficient on the R&D variable would proxy partly for exogenous technological progress. Satisfactory instrumental variables to avoid this problem may not be available." Few studies have estimated the return to R&D using instrumental variables; Jaffe (1986) is perhaps the major exception. Lewbel (1997) uses instrumental variables in a study of the effect of R&D on patents.

sharply different in industry data than in firm data. If satellite accounts for R&D are to include the economic impact of spillovers, the key channel through which R&D affects productivity and output growth, national income accountants will have to move beyond asset effects to examine spillovers, using alternative values to measure central concepts such as depreciation, lags, and the rate of return.³¹

Considerable evidence describes leakage of R&D from firms to the broader industry. Mansfield (1985) showed information about new technology leaks out to rival firms within a year or two. Caballero and Jaffe (1993) similarly find rapid leakage. Jaffe (1986) found the pool of nearby research affected technological progress (the number of patents) more strongly than it affects private returns (the profits or market value of firms).³²

Nelson's argument that R&D reflects the presence of collateral technological opportunities applies with particular force at the industry level, where major differences in technological possibilities are most visible. For example, no one would reasonably

³¹ Though spillovers are important within an industry, many spillovers of course take place in a broader context, to other industries, within a national economy, or internationally.

In practice, it is often difficult to determine the rate of depreciation and the lag until research becomes effective. For example, a detailed study of 133 firms (Griliches and Mairesse (1984)), which was designed to examine the effects of various ways of defining and measuring physical and R&D capital (p. 347), was not able to establish definitive estimates of the depreciation rate and the lag (pp. 372-373).

Market share structure also affects the difference between firm and industry results. Blundell, Griffith, and Van Reenen (1999) show firms with higher market share tend to have a greater number of innovations and patents, and the effect of innovation on stock market value is greater for when firms with larger market share. Such results show firms with larger market share are able to appropriate a larger proportion of the industry gains to R&D.

³² Baumol (2000) emphasizes that the different firms within an industry often use R&D to compete on the basis of product quality, and that the cumulative nature of progress ensures that product quality improves greatly over time. As Sena (2004, p. F314) summarizes Baumol's argument, "spillovers and voluntary dissemination allow technological knowledge to spread in the economy so as to increase productivity of other firms and this way guarantee the continuous growth and improvement in living standards Western economies have experienced in the last two centuries."

expect productivity growth to become as rapid in lumber as it is in computers if R&D intensity in lumber were raised to the levels prevalent for computers. Yet, that is what the usual cross-industry regressions assume. Nelson recommended that the return to research should be determined not from regressions across firms or industries, but instead from careful study of the returns to individual innovations. Subsection II.5 examines rates of return obtained from innovation studies.

II.3 Estimates of Indirect Returns to R&D – Downstream Analyses of R&D.

The notion of complementary investments has been important for a long time in the understanding of indirect returns to R&D. For example, it has long been the case that, when a new computer is introduced, the purchasing firm must deploy considerable resources to use the new equipment effectively. Bresnahan and Greenstein (1996) describe such complementary investments as coinvention, and Bresnahan (1999) emphasized firms must adjust their organizational structure and their work processes, which is a long and slow procedure, to take full advantage of the new technologies.

In recent years, economists have begun to pay far more attention to the idea that new technology requires very substantial complementary investments. Brynjolfsson, Hitt, and Yang (2002) argue that such complementary investments are very large. They show that financial markets valued a dollar's worth of computer capital as ten dollars. Other kinds of investments do not have such large effects in the financial markets. The authors argue that the great effect which computers have on firm financial value occurs because large complementary investments, in such areas as software, organizational

redesign, and worker training, are typically made together with the purchase of computers. As they point out, such adjustments eventually extend to many fundamental characteristics of the firm, such as the extent of centralization or decentralization of decisions, patterns of work flow, work compensation and incentive patterns, relationships with customers and suppliers, and even corporate culture. Expenditures on improving business practices of this type are typically expensed. As Brynjolfsson, Hitt, and Yang (2002) argue, if expenditures in these areas increase substantially with computer spending, and are a source of unmeasured investment and value to the firm, then the observed increase in firm value with computers can provide an estimate of the associate increased investment in intangibles. Bresnahan (1999) also provides a very useful discussion of the ways in which organizations have to restructure in order to use new technology more effectively. Finally, Brynjolfsson and Hitt (2003) have recently reported that the observed "return" to computer investments, at the firm level, grows steadily over time, which they interpret as ever increasing investment in complementary assets.33

The clear importance of complementary assets makes it far more difficult to determine the appropriate rate of return to R&D. If downstream complementary investment is substantial, it becomes very difficult to determine how much of the

³³ Bloom, Sudun, and Van Reenan (2006) provide important support for the idea that corporate knowledge of information technology applications is central by showing that, in Great Britain, U.S. multinationals are more productive than non-U.S.multinationals or domestic firms, and that these productivity advantages are associated with higher returns to information technology. Furthermore, these high returns to information technology are greatest in sectors such as trade and finance where the U.S. productivity advantage over Europe, as measured within industry data, has grown most rapidly,

observed downstream returns are attributable to indirect R&D and how much is due to the many different forms of complementary investment.

Does indirect R&D reflect the presence of measurement error

Griliches has pointed out (Griliches (1992)) that observed indirect returns to R&D could represent either measurement error or true spillovers, which are ideas borrowed by one research team from another.³⁴ In practice, as many authors have pointed out, the line between measurement error in prices and genuine technology transfer is difficult to distinguish. For example, when an aircraft manufacturer prepares a new plane, they consult carefully with customers, adapt the plans so as to ensure consumer satisfaction, and spend considerable resources on training airlines to use the new aircraft. All these mechanisms create genuine technology transfer. When statisticians create a price for aircraft, no matter how detailed the specification for an airplane is, such detailed characteristics of a sales package are not likely to be included. A single transaction can therefore represent both technology transfer and measurement error in prices.

Nevertheless, there is important evidence that a lot of the indirect returns to R&D reflect measurement error. Griliches and Lichtenberg (1984a) present a model in which errors in the price of equipment or the price of materials are related to the R&D intensity of the sector supplying these goods; the underlying logic of their model implies that the indirect effects of R&D primarily represent measurement error. In addition, Gordon (1990) demonstrated that equipment prices for highly research intensive capital goods

³⁴ Measurement error occurs if the price of output, in an industry such as computers, does not fully reflect improvements in the quality of output. The purchaser then obtains greater quality benefits than the measures of price suggest.

contain substantial measurement error. More recently, Greenwood, Hercowitz, and Krusell (1997) argued that a declining relative price of equipment, heavily influenced by Gordon's estimates of the quality-corrected price of equipment, has been an important ingredient in growth.

If observed spillovers associated with indirect R&D primarily reflect measurement error in prices rather than the transmission of ideas, these returns cannot be used as a contributor to productivity growth and as a source of output growth. As Jorgenson (1966) showed long ago, if a correction is made to increase the growth of output in a poorly measured capital goods sector, this adjustment will increase output and multifactor growth, but the corresponding increase in higher quality capital input, purchased by customers, will reduce multifactor productivity. The two effects roughly offset each other, and there is little net effect on multifactor productivity growth. It is therefore an important issue whether the spillovers associated with indirect R&D reflect only measurement error. In the likely instance in which a large proportion of indirect R&D spillovers reflect measurement error, this portion of the spillover cannot be used to explain multifactor productivity growth. When Fraumeni-Okubo (2005, Table 1) include indirect rates of return in their estimates of the overall return to R&D, they implicitly assume that none of the spillovers associated with estimates of the social return to R&D reflect measurement error, which is not likely to be correct (Gordon (1990)). Their reference to the possibility of measurement error (footnote 9) does not sufficiently warn the reader that, if measurement error is important, most of the estimates of the indirect return to R&D should not be used to help determine the overall return to R&D.

In view of these considerations, estimates of indirect returns are subject to important limitations. Heavy downstream complementary investments distort the picture, and there is evidence of considerable measurement error, which implies that any substantial net effect on productivity is questionable. In view of these difficulties, one has to be extremely careful about adopting estimates of the rate of return to R&D which rely heavily on estimated indirect returns.

II.4. The Spillover of Ideas.

Early work on R&D concentrated on the returns to firms and industries. However, more recently, spurred by comments in Griliches (1979; 1992), attention shifted towards true spillovers of ideas. Many studies have used detailed information from patent records to trace the evolution of ideas.

Jaffe (1986; 1988) examined the effects of knowledge spillovers between firms, where intrafirm spillovers depended on the similarity of firm's patent portfolios. Not surprisingly, the spillover of nearby R&D had a greater impact on technical progress, as measured by patents, than on profits or stock market value. Verspagen (1997a; 1997b) and Frantzen (2002) used detailed information from patent records to describe knowledge spillovers between countries and industries.³⁵

³⁵ These studies examine the effect of R&D in the domestic industry, R&D in related industries, foreign R&D in the same industry, and foreign R&D in associated industries. Though an overall R&D effect is clear, the more detailed measures often face collinearity difficulties (Verspagen (1997b).

In other work, Park (2004) examines international and intersectoral spillovers of R&D, and concentrates on spillovers to production outside manufacturing, an important issue which deserves further consideration. Moretti (2004) examines human capital spillovers within metropolitan areas using several measures of technical connection between different industries obtained from the R&D literature. He finds

Jaffe and Trajtenberg (2002) carried out extremely comprehensive work which provides detailed information on each individual U.S. patent from 1963 to 1999. Their work emphasizes the forward citations gained by each patent.³⁶ In conjunction with evidence from Trajtenberg (1990), Harhoff, Narin, Scherer, and Vopel (1999), and especially from Hall, Jaffe, and Trajtenberg (2005), this line of work shows how important forward citations are.³⁷

Jaffe and Trajtenberg (2002) include a diskette, which makes their detailed data on United States patents easily accessible. Their generosity in making these data available has already led to a broad literature (Peri (2005); AlAzzawi (2004)) which has begun to make important contributions to a variety of issues.³⁸ Somewhat fortuitously for the purposes of the present discussion, the release of the Jaffe-Trajtenberg data provides an example of the vast downstream spillovers potentially associated with important technological progress.³⁹

Though much of the literature on spillovers concentrates on spillovers across industries or countries, work has also begun to examine spillovers in many other

input-output connections and Jaffe's measure of the similarity between industry technologies work better than interindustry patent citations.

³⁶ Forward citations are citations a patent earns in the future. Backward citations are citations a patent application makes to prior work.

³⁷ Lanjouw and Schankerman (2004) consider the relative importance of several indicators of patent quality, including forward cites, backward cites, and the number of claims made in each patent. All three of these effects have some importance.

³⁸ Almeida (2004) analyzes highly detailed information on the identity of individuals who have filed important patents to study the effect of labor mobility on technical change. Almeida's work on individual inventors illustrates the rich potential of the more detailed information included in each patent.

³⁹ The availability of the Jaffe-Trajtenberg data is an example of within industry spillover, if the relevant industry is defined to be economic research. Clearly, the importance of within industry spillovers depends on the definition of an industry.

circumstances. Jaffe, Fogarty, and Banks (1998) studied how research carried out by NASA spilled over to industry. Adams, Chiang and Jensen (2003) examined how research conducted in Federal research laboratories, which accounts for 14 percent of total United States research expenditures, spilled over to private firms. Building on Mansfield (1995), Adams (2004) investigated how research conducted in specific university departments affected industrial users. Darby and Zucker (2003) described how star scientists have been influential in transferring ideas to biotechnology firms.⁴⁰ It is useful to understand the transfer of ideas through all these different types of linkages. These articles show that, beyond the firm to firm transfers that comprise the core of the R&D literature, substantial technology is also transferred from government or universities to private firms.

IV.5 Evidence on Private and Social Returns from Studies of Individual Innovations.

In view of the difficulties associated with estimates of direct and indirect returns (Subsections IV.2 and IV.3 above), it is tempting to follow Nelson (1988) and obtain private and social returns to R&D from studies of individual innovations. The major article which examines individual innovations is Mansfield <u>et al.</u> (1977).

There are two problems with relying on Mansfield's results. First, the study included only 17 innovations. However, the National Science Foundation soon replicated Mansfield's work (Tewksbury, Crandall, and Crane (1980); Nathan Associates (1978)). The three studies together consider 57 innovations.

⁴⁰ See also Zucker, Darby, and Armstrong (1998).

Second, Mansfield <u>et al.</u> (1977) emphasize the median rate of return to individual research projects (25 percent private returns and 56 percent social returns). If one is interested in the return to a typical scientific project, the median rate of return is relevant. However, to understand the effect of R&D on productivity and output growth, the mean project return is relevant. As Scherer (1999, Chapter 5) shows, most research projects are not particularly successful, but a few very large successes drive total returns. If one looks only at median returns, the very large effects which the most successful projects have on output and productivity growth will be missed.

In all 57 innovations, the median private return to research is estimated to be 28 percent, and the median social return is 71 percent. This information suggests social returns exceed private returns by 43 percent. The three innovation studies include 7 observations where the private return is negative, and 5 where social returns are negative. The studies do not report specific losses for failing innovations. Assuming negative returns are losses of 50 percent, average private returns are 32 percent, and average social returns, boosted by 11 innovations where the reported social return is greater than 150 percent, are 87 percent.⁴¹

There is one potential problem with these innovation data. Scherer (1999) showed most innovations are typically not profitable. However, the private rate of return was negative for only 7 of the 57 innovations. The low incidence of negative returns suggests that the innovations considered may not be representative.

⁴¹ Social returns of the magnitudes mentioned in this paragraph imply that private returns represent considerably less than half of the total social return.

In summary, Mohnen (1996, p. 50) concluded, based on a broad overview of the R&D literature, that "On average, the social return to R&D exceeds the private return by 50 to 100 per cent." The innovation studies provide important further evidence that social returns are very large and exceed private returns by a substantial margin.⁴²

IV.6 The Fraumeni-Okubo Summary of Evidence on the Rate of Return to R&D.

Fraumeni and Okubo (2005, Table 1) recently selected nine studies to determine the rate of return to privately funded R&D.

Author (year)	Private return	Social return
Sveikauskas (1981)	7-25 %	50 %
Bernstein-Nadiri (1988)	10-27	11-111
Bernstein-Nadiri (1991)	15-28	20-110
Nadiri (1993)	20-30	50
Mansfield (1977)	25 (median)	56 (median)
Goto-Suzuki (1989)	26	80
Terleckyj (1974)	29	48-78
Scherer (1982, 1984)	29-43	64-147

Table 1. Evidence on the Rate of Return to Private R&D. (from Fraumeni and Okubo)

Five of the nine papers (Sveikauskas (1981), Goto-Suzuki (1989), Terleckyj (1974), and Scherer (1982, 1984)) are industry studies, and report the return to research conducted "Within" the industry and the return to R&D "From Outside", that is R&D carried out by other industries.⁴³ Although some text (Goto and Suzuki (1989, p. 561)

⁴² Wolff (1997) concludes the social return is 53 percent and the private return is about 10 to 12.5 percent, which also implies social returns exceed private returns by 40 percent.

⁴³ Section III of Goto-Suzuki reports estimates of the private return to firms, but Sections IV and V of their paper, which deal with social returns, are based on industry data.

interprets these coefficients as private and social returns, "private" returns here reflect industry returns, not firm returns. Industry returns correspond to firm returns only if there are no spillovers within an industry.⁴⁴

Bernstein and Nadiri's (1988, 1991) work is slightly different. They analyze industry data, but remove the capital (1988, page 431) and labor and materials (1991, page 37) used in research from the BLS or NIPA factor inputs typically used to study U.S. productivity. By making this adjustment, they are able to estimate total returns to R&D rather than just R&D spillovers.⁴⁵ However, the coefficients they estimate at the industry level show the sum of the private return to the firm undertaking R&D investment <u>plus</u> spillovers to other firms in the industry. These returns are not the same as "private returns", the total returns to an individual firm undertaking research, as described in Mansfield <u>et al</u>. (1977).

⁴⁴ Fraumeni and Okubo (2005, Table 1) list the private and social returns to R&D reported in these studies. They attribute the numbers to a Council of Economic Advisers file, which no longer exists on the Web. Joseph Stiglitz, then Chairman of the Council, distributed a handout based on this file during a talk in Washington. Fraumeni and Okubo report the Council's summary accurately. However, the columns of the Council table were not properly labeled. As discussed in the text, only Mansfield <u>et al.</u> (1977) reports a private rate of return, in the traditional sense of the total returns to a private firm. Some of the ambiguity comes from Griliches (1992, Table 1, part II), which combines information on the private and social return to R&D with parallel information on the direct and indirect return within a single table. The text of that paper (page S43) comments that the estimated social rates of return look surprisingly uniform, but does not emphasize that many of the rates of return reported in the left column of Table 1 (listed as Within and From Outside, respectively) include elements of social returns (the within industry spillover, and the excess return, which exceeds the costs of R&D) and does not represent the traditional private return to firms. (Griliches (2000, pp. 60, 61, and 70) shows returns to R&D are considerably greater in industry data than in firm data.)

⁴⁵ Current BEA work on R&D (Schultz (2006)) now removes factor expenditures on R&D from factor inputs, using recently expanded NSF data on the composition of research costs in each industry. This important advance permits BEA to estimate the total return to research, rather than just the spillover component of returns. For a discussion of why standard procedures estimate only the spillover effect see U.S. Bureau of Labor Statistics (1989, page 36).

Nadiri (1993) summarizes results from many different studies, on both a firm and industry basis. The only study on the Fraumeni-Okubo list which actually estimates private and social returns is Mansfield <u>et al</u>. (1977), which is fundamentally different because it measures the return to individual innovations rather than firm and industry regressions. Except for Mansfield, the studies cited do not provide information on the private and social rates of return to R&D.

In view of these many difficulties, what R&D expenditures should be included as investment and what rate of return should be attributed to them? First, in industry, all privately financed research should clearly be counted as investment. However, the evidence indicates federally financed research in industry, on net, does not have a positive return, so such expenditures probably do not qualify as investment. Second, because of the issues of relevance to growth and double counting discussed in Section I, only a small proportion of the research financed by government or universities should be classified as investment.

Third, the studies Fraumeni-Okubo (2005) list in their Table 1 are far too heavily weighted towards industry studies of direct and indirect returns. Such estimates depend much too heavily on indirect returns to R&D, which are unreliable because they depend on the amount of complementary investment, and at least partially reflect measurement error, so that the estimated indirect effects are probably not a genuine effect on productivity. Fourth, because of the many difficulties with the regression estimates it is probably useful to emphasize rates of return obtained from the innovation studies, which also indicate that social returns are very large and considerably exceed private returns. On balance, private returns of 25 percent and social returns of about 65 percent, which more than double the private return, seem reasonable. However, these extremely high returns are relevant <u>only</u> for privately financed research.

III. <u>R&D Stocks on an Industry Basis</u>.

In the growth accounting literature, residual income, which remains after all other factors of production are compensated, is generally attributed to capital. The standard treatment of capital in growth accounting, due to Jorgenson and Griliches (1967), first removes payments to labor and materials from income, and then, using the Hall-Jorgenson (1967) procedure, assigns the sum which remains, property income, to specific capital assets. These procedures first allow for depreciation, taxes, and other costs of capital, and then allocate the rest of capital income to the internal rate of return, or capital profit.

Much evidence suggests that observed profit income is not just a return to capital, but also reflects returns to research and development and other intangible assets. For example, Grabowski and Mueller (1978) examine a sample of 86 firms which conduct at least some R&D. They report (p. 342) the overall variance in profit rates was reduced approximately in half when R&D was capitalized rather than expensed. They conclude that the after-tax return to R&D is between 15 and 20 percent, considerably above the rate of return earned on other assets, but comparable to the returns reported in Mansfield et al. (1977). The Grabowski and Mueller study clearly shows that some of the income typically included in the return to capital actually represents a return to R&D.

Recently, economists have discussed investments in many types of intangibles which extend far beyond R&D. Corrado, Hulten and Sichel (2003) estimate that expenditures on intangibles in the United States were \$1221 billion, annually, in the United States from 1998 to 2000. R&D expenditures, as defined by the National Science Foundation, were \$185 billion, or only about 15 percent of the total. Investments in knowledge in other areas, such as finance, movies, and books, which the NSF does not count as research because they are not based on an explicit scientific or technological field, were estimated to be \$199 billion.⁴⁶ Among other items, computer software accounted for \$154 billion, advertising for \$254 billion, employee training for \$123 billion, and management improvement for \$306 billion. R&D accounts for only a small share of the amount firms spent, beyond capital expenditures, to increase future output.⁴⁷

The Multifactor Productivity program of the Bureau of Labor Statistics currently includes plant, equipment, inventories, land, and software as capital assets, and calculates the internal rate of return, service price, and income paid for each asset. A separate internal rate of return and industry-specific asset service prices are calculated for 56

⁴⁶ The National Science Foundation publication <u>Research and Development in Industry</u> does not include R&D in the social sciences or psychology, so research in financial economics does not appear in these data. ⁴⁷ Many of the assets Corrado, Hulten, and Sichel (2003) consider depreciate considerably more rapidly

than tangible capital and R&D, so the proposed new items account for a smaller proportion of the overall capital stock than these figures on investment indicate.

different industries. If R&D is to be treated as an investment, it would be natural to expand the BLS calculations, on an industry basis, to include R&D as a further asset.⁴⁸

Unfortunately, it is difficult to determine the relevant R&D stock within each industry in the United States. As Scherer (2003) discusses, in the U.S. most data on R&D are collected at the firm level. Many large firms invest substantial sums in R&D. However, these firms frequently operate in many different industries, and it is not clear how much of each firm's total research portfolio supports production in any particular industry.⁴⁹ As an illustration of the difficulties involved, Wolfe (2003) reports the U.S. National Science Foundation assigns large amounts of R&D to the trade industry, even though relatively little research actually takes place in trade. The problem occurs because all research conducted by a firm is assigned to the one industry in which its payroll is greatest. If a pharmaceutical firm has many sales representatives, its research efforts could well be assigned to trade, even though all the research actually takes place in pharmaceuticals.

Because these procedures assign all research conducted by a firm to a single industry, current National Science Foundation data are not accurate enough to create a reliable measure of R&D stocks within each of the BLS industries.⁵⁰ In my judgement, it

⁴⁸ The BLS calculates a separate internal rate of return for each of 56 industries included in the nonfarm private business sector. To extend these calculations to R&D requires R&D stocks for each of these 56 industries.

⁴⁹ Brown, Plewes, and Gerstein (2004) discuss the limitations of available measures of R&D in specific industries in considerable detail.

⁵⁰ The Bureau of Economic Analysis faces similar issues in establishing industry R&D stocks. Carson, Grimm, and Moylan (1994), and much of the rest of the R&D literature, has used the existing NSF data on industry R&D expenditures to create industry R&D stocks.

is necessary to modify the NSF data on industry R&D expenditures considerably before they can provide a sound measure of R&D assets in each industry.

A potential method of assigning R&D spending conducted by individual firms to specific industries could follow the well-established procedures which the Bureau of Economic Analysis currently uses to allocate profits earned by firms to specific industries. The BEA takes Internal Revenue Service measures of the profits earned by firms classified in each industry, and allocates these profits to establishments in particular industries on the basis of data showing the industries in which these firms actually employ workers, as determined from the Enterprise Statistics.

Similar procedures are likely to improve the accuracy of measures of the R&D actually committed to each industry. A separate allocation can be conducted for every firm, allocating that firm's research in proportion to its own industry distribution of employment or payroll. Furthermore, firm R&D could be assigned to particular industries not only on the basis of employment or payroll, but also on the basis of the intensity of scientific and technical employment within each industry. Through such a methodology, trade, where scientific input is typically low, would for that reason be assigned less of a firm's research than pharmaceuticals, where scientific input is typically high.⁵¹ A simple allocation procedure of this type is likely to provide better estimates of

⁵¹ As an example, each firm's proportion of its work force in a given industry, s_i , could be multiplied by the proportion of technological employment in that industry in general, t_i . Then, that firm's R&D expenditures could be allocated to specific industries in proportion to $s_i t_i$ as a share of the firm's total $s_i t_i$ across all i industries. If R&D is allocated by payroll, in the second stage industry totals would have to be adjusted to reflect the technological employment/payroll or technological payroll/payroll ratios.

In a slightly different, but probably preferable, methodology, each firm's research expenditures can be regressed on its payroll in each relevant industry. The resulting industry coefficients then provide an

the R&D actually occurring in each industry than those obtained by assigning each firm's R&D solely to one industry. Furthermore, since a large number of firms contribute to production in any given industry, the proposed alternative procedure is less likely to have problems with disclosure limitations.⁵² The procedures proposed here would probably make it possible to publish R&D estimates for a larger number of industries, which would help overcome one of the major limitations of current NSF industry R&D data.⁵³

Once the distribution of R&D across industries is better understood, more reliable information on the industries in which R&D actually occurs can in turn be used to determine the industries in which R&D is actually utilized. The literature (Scherer (1982, 1984); Verspagen (1997a; 1997b); Moretti (2004)) has used input-out relationships, information from patent applications, and careful analysis of individual patents to determine the specific industries in which R&D spillovers occur.

The Bureau of Labor Statistics currently uses an interesting framework to understand the role of R&D spillovers in the aggregate economy. Spillovers do not provide income to any particular asset, and are not included within the data on factor inputs and factor earnings which determine the portion of growth attributable to specific inputs. Instead, aggregate spillovers are included "below the line", as one of the determinants of observed

estimate of research intensity in each industry. It would be of interest to examine how closely such industry coefficients match actual measures of scientific or technical employment in each industry. The procedure suggested in the present paragraph determines research intensity in each industry from a statistical procedure, rather than by assumption, and utilizes the highly detailed data available for each firm. Because more data are utilized, this variant of the methodology seems likely to provide estimates of research intensity for a larger number of industries.

⁵² For example, if each firm produces output in five industries, the data cell for a typical industry would contain information from five times as many firms.

⁵³ I plan to construct and analyze alternative measures of industry R&D, prepared along the lines discussed above.

multifactor productivity growth. Such procedures are helpful in evaluating the role of spillovers in national data. However, further information on the industry distribution of spillovers is necessary to extend this approach to specific industries.

IV. International R&D Spillovers.

As evidence presented below indicates, even for the United States, a very large and highly advanced economy, foreign R&D plays an important role in technical progress. Most of the other countries in the world, which are much smaller and less technically progressive, are far more dependent on foreign R&D. Consequently, most of the statistical agencies in the world will have to understand and evaluate how foreign R&D affects their economies in order to include the full effect of R&D in their accounts.

R&D is heavily concentrated in a few developed countries, essentially in the OECD.⁵⁴ Even within the OECD, a few relatively large countries, like Germany, Japan, and the U.S., account for a large proportion of R&D.⁵⁵ Once R&D is included in the national income accounts, economists will have to provide the underlying information on how R&D in the leading countries affects production in small rich countries or in the developing world.

⁵⁴ The OECD countries had a particularly large share of world R&D in the 1970s, 1980s, and 1990s. However, in very recent years R&D has increased sharply in several countries outside the OECD, notably China and India.

⁵⁵ For example, in 1995 the United States, Japan, and Germany accounted for more than two-thirds of OECD R&D expenditures, and the G7 countries together accounted for about 85 percent of the total. Data are from the OECD Science, Technology, and Industry Scoreboard: 2003.

There are many potential channels through which the undoubted technological advantages of the most advanced countries can be transmitted to less wealthy nations. One branch of the literature emphasizes the mere presence of international trade (Coe and Helpman (1995)). Other analyses emphasize the presence of multinational companies (Gorg and Strobl (2001)) or the role of foreign domestic investment (Borensztein, De Gregorio, and Lee (1998), Aitken and Harrison (1999)).⁵⁶ The transmission of ideas (AlAzzawi (2004)), through such mechanisms as the scientific or technological literatures, is another potentially important pathway. Of course, in actual practice many or all of these potential channels of transmission are likely to be influential. Much work still remains to be done to sort out how these potential influences interact and to understand the particular circumstances under which each is important.⁵⁷

Even among rich countries, there is a distinction between the largest countries, such as the U.S., Germany, and Japan, which conduct large amounts of research, and small countries which perform much less research. Within the research leaders, spillovers within an industry and across industries tend to be very important. In smaller countries, many of the equivalent effects come into existence through foreign trade spillovers.⁵⁸

⁵⁶ Although much of the literature concentrates on whether technology from multinationals spills over to domestic production, multinationals of course make a contribution if they produce at a higher level of productivity than domestic firms.

⁵⁷ Recent studies have begun to investigate links between different facets of technology transfer. For example, AlAzzawi (2004) examines the extent to which spillovers of knowledge, as measured by patent citations, are associated with foreign domestic investment. Branstetter (2000) examines whether Japanese investment in the United States is a channel of knowledge spillovers to Japanese firms.

⁵⁸ Tables 4 and 5 of Coe and Helpman (1995) support the statements made in this paragraph. As in most estimates of this type, spillovers from the U.S., Germany, and Japan are larger than those from other countries. Coe-Helpman also find domestic R&D is more important than foreign R&D in large countries, but foreign R&D is more important than domestic R&D in most small countries.

One branch of the empirical literature examines research spillovers among advanced (OECD) nations. Coe and Helpman (1995) emphasize the volume of foreign trade as a key to research spillovers. Their work also includes imports as a proportion of GDP, which permits technology transfer to vary with the importance of trade. In later work, Lichtenberg and van Pottelsberge de la Potterie (1998) suggest a better measure of R&D flows, and Funk (2001) and Kao <u>et al</u>. (1999) clarify the econometric issues involved. Despite these improvements, Coe-Helpman (1995) remains the classic article which demonstrates the potential importance of international R&D spillovers.

Xu and Wang (1999) is a particularly useful study of R&D spillovers within the OECD. They show R&D in capital goods affects productivity growth, but R&D contained in other forms of trade does not. Xu-Wang include domestic R&D stocks as well as a proxy for the extent of the disembodied flow of information (a distance weighted, or unweighted, measure of other countries R&D stocks) in addition to the R&D contained in capital goods. The broad range of variables allows them to determine the importance of R&D transmitted through each of these channels. Xu and Wang (1999) is one of the most insightful and helpful contributions in the literature because they report estimates of returns obtained through each mechanism.⁵⁹

⁵⁹ Verspagen (1997b) similarly concludes that R&D spillovers embodied in purchased inputs and knowledge floating freely across international borders both play a role in technology transfer across countries. Like Xu and Wang (1999), Verspagen (1997b) shows R&D spillovers from the large technology leaders have the greatest impact on other nations.

Griffith, Redding, and Van Reenen (2004) integrate R&D with international technological convergence. They demonstrate that, in additional to the usual R&D effect (observed at the industry level), there is convergence (in the sense that country-industry observations at a lower level of productivity tend to catch up), and that R&D permits the successful absorption of foreign technology. Keller (2002) studies R&D spillovers from the largest countries to smaller OECD nations.

Another strand of the literature considers R&D transfers to countries outside the OECD, typically developing nations which conduct very little R&D. Coe, Helpman, and Hoffmaister (1997) is the original contribution in this direction. Schiff, Wang, and Olarreaga (2002) and Connolly (2003) conduct more recent work. Schiff, Wang, and Olarreaga (2002) find R&D imported in capital or materials from Northern countries increases productivity in Southern industries, especially in R&D intensive industries.

Many studies of technology transmission to developing countries (Borensztein, De Gregorio, and Lee (1998); Connolly (2003)) examine data for the aggregate economy. More information could be brought into play if economists constructed more reliable measures of industry productivity growth in different countries.⁶⁰

The effect of technology transfer on developing countries will eventually have to be understood on the basis of highly detailed information on firms and establishments. Muendler (2004) recently studied the type of information which is needed.⁶¹ His work on

⁶⁰ Two recent studies illustrate both the potential of international comparisons based on detailed industry data and the need for better measures of productivity. Savvides and Zachariadis (2004) examine the effects of research contained in foreign capital goods, the amount of capital goods imports, and foreign investment on productivity growth in manufacturing. However, economy wide deflators, rather than data specific to manufacturing, are used to measure real ouput. Similarly, Schiff, Wang and Olarreaga (2002) examine how R&D imports affect productivity growth in 16 manufacturing industries in 25 developing countries. Output data are expressed in U.S. dollars and deflated by the U.S. GDP deflator. It is questionable whether this procedure generates productivity measures which are as reliable as studies which use industry specific output price deflators for each country.

The International Comparisons of Output and Productivity (ICOP) project at the University of Groningen has, for many years, measured international productivity differences based on detailed information on unit values in different countries.

⁶¹ Hallak and Levinsohn (2004) similarly recommend the use of highly detailed micro data, rather than aggregate country data, in the analysis of the closely related issue of whether increased amounts of foreign trade bring a positive effect to the growth rate. As in the present review, they recommend greater emphasis on the analysis of, and discrimination between, the many different channels through which growth effects may operate. Hallak-Levinsohn also argue that different channels are likely to be influential in different circumstances. A recent review (Keller (2004)) also emphasizes the need for detailed data, as well as the usefulness of case study evidence.

productivity in Brazilian firms included measures of the foreign equipment and materials available to each firm. He concluded foreign import competition was more important than foreign technology.⁶² If further work, especially with detailed establishment data on other countries, confirms these results, technology may be less important than previous discussion has suggested. In that event, greater weight will have to be attached to product market competition, as emphasized in Parente and Prescott (1994; 1999) and Lewis (2004). It is worthwhile to be reminded that technology transfer can be much more effective when other economic conditions are supportive.⁶³

On balance, there is widespread international transmission of knowledge associated both with the imports of R&D-intensive capital equipment (Xu and Wang (1999), Coe, Helpman, and Hoffmaister (1997), and Caselli and Wilson (2004)) as well as R&D included in materials as well as capital equipment (Schiff, Wang and Olarreaga (2002)). This international transmission of R&D typically has a favorable effect on productivity growth. One interesting study finds strong benefits associated with R&D intensive materials (Lichtenberg and Virabhak (2002)).

In addition to Muendler (2004), several further studies examine detailed data on production in developing nations. Basant and Fikkert (1996) analyze panel data for Indian firms between 1974 and 1982. India appears to be a developing country in which many firms conduct their own R&D. They emphasize foreign purchases of technology, as well as the impact of domestic R&D, and construct a foreign R&D spillover measure by weighting R&D in eight wealthy economies by each country's relevance to India, as determined by patents granted in India.

In a production function study, Bartel and Harrison (2005) compare the productivity of public sector and private sector production in Indonesia. Among other topics, they consider the effect which foreign ownership has on productivity. Blalock and Veloso (2006) examine the effect of imports on productivity in Indonesian data. All this work illustrates the usefulness of highly detailed datasets describing productivity growth over time. See also Haskel, Pereira, and Slaughter (2007) for the United Kingdom.

⁶² In further evidence supporting the Parente-Prescott and Lewis view, Schiff and Wang (2004) show that, in North-South trade, openness to trade, as measured by the amount of trade, has a greater impact on productivity growth than the knowledge (R&D) content of a unit of trade does.

⁶³ Some scholars emphasize countries gradually have to develop the capacity to use modern technology. Nelson and Pack (1999) present such a viewpoint, and compare the results with growth accounting.

There is also substantial transmission of knowledge not specifically associated with factor inputs (Xu and Wang (1999)). Such disembodied knowledge flows are often highly localized (Peri (2005), Maursith and Verspagen (2002)). The importance of distance may change greatly over short periods of time (Keller (2002)).

A large advanced economy such as the U.S. generates most of its own technical progress. Foreign R&D could be most influential if it allowed U.S. research to improve more rapidly. However, it is difficult to evaluate the importance of foreign ideas. As U.S. Bureau of Labor Statistics (1989, page 6) remarked:

"A further issue is whether foreign research investment should be included. Clearly, as foreign technological levels have approached U.S. levels and as American multinational corporations have conducted more research abroad, foreign R&D has become more relevant to U.S. firms. However, the relative weight to be attached to a unit of foreign research and how this may have changed over time are unclear."

Data on patent citations now make it possible to determine the relative importance of foreign research, and how this changes over time. Consider all U.S. patents granted to inventors located in the United States. Such patents arguably provide a good description of U.S. technical progress each year. Table 2 shows how often these U.S. patents cite U.S. and foreign research.⁶⁴

⁶⁴ I thank Anastasiya Osborne for programming the large data bases containing these patent data. If one looks only at citations of foreign work patented in the U.S., it is necessary to examine how much foreign research is patented in the U.S., and how this proportion changes over time (Eaton and Kortum (1996; 1999)). However, the data reported in the text include all citations made by U.S. patents, regardless of whether the cited patents are U.S. or foreign. If each citation made by a U.S. patent is assumed to be of equal value, a simple count of the domestic or foreign origin of citations provides direct information on the relative importance of domestic and foreign research. Because large numbers of citations are observed each year, differences in the importance of individual citations should cancel out.

The available data report the country of cited patents only for patents granted in 1963 or later. We estimate the country origin for patents granted prior to 1963 on the basis of time trends in later years. It is

1976	.301	1981	.366	1986	.435	1991	.490	1996	.505
1977	.314	1982	.389	1987	.447	1992	.503	1997	.510
1978	.327	1983	.399	1988	.455	1993	.510	1998	.497
1979	.338	1984	.405	1989	.470	1994	.521	1999	.506
1980	.356	1985	.426	1990	.473	1995	.552	2000	.525

Table 2 Citations of Foreign Patents, in Proportion to Citations of U.S. Patents 1976-2000 (for utility patents, year refers to year of application)

Source: computed from files of individual patents, as provided by the U.S. Government Patent and Trademark Office.

These data suggest that in 1976, in addition to knowledge obtained from U.S. R&D, the United States had access to 30 percent further knowledge from foreign research.⁶⁵ By 2000, the increment of foreign knowledge increased to 53 percent.

U.S. productivity growth accelerated sharply in the late 1990's, and the productivity recovery was sharper in the U.S. than abroad. Table 2 shows the proportion of citations of U.S. work declined steadily from 1976 to 1995, but this trend then reversed. Patent citations suggest U.S. technology turned a central corner in 1995; the timing is remarkably similar to the year in which U.S. productivity growth accelerated.

important not to omit cited patents occurring before 1963, which are overwhelmingly of U.S. origin. Early versions of this paper reported a greater proportion of foreign citations, because citations to patents occurring prior to 1963 were not yet included.

⁶⁵ These calculations assume that each citation, whether domestic or foreign, contributes the same amount of knowledge to U.S. innovation. The proportion of foreign citations is greater than found in many other studies because citations of patents granted outside the U.S are included. For example, in 1976 there were 34984 citations of U.S. patents granted to foreign inventors, and 17370 further citations of foreign patents.

If we assume foreign R&D increased U.S. effective R&D stocks by the ratios given in Table 2, and foreign knowledge earned the same 30 percent rate of return, foreign R&D contributed 0.1 percent a year to 1947-2002 U.S. productivity growth.⁶⁶

A major unresolved question is what happens to returns to R&D in the U.S. and other advanced countries as further nations, such as China and India, expand their capability to do R&D. So far, most studies conclude R&D in one country increases productivity in other countries. However, Luintel and Khan (2004) report foreign R&D reduces productivity in the United States. The effect of international competition in technology on the returns from R&D is a crucial question which requires much further analysis.^{67 68}

V. Conclusions.

Recent decisions to include research and development expenditures in the System of National Accounts raise important new questions for statisticians in developed and developing countries. Many of the benefits of R&D are not obtained by the original investors, but instead leak out to other firms, customers and even other nations. Because the asset and spillover influences of R&D have different time horizons, it is useful to establish separate "national R&D stocks" for each purpose.

⁶⁶ Under these assumptions, the 1948-2002 R&D effect increases from .17 percent a year (with the usual BLS estimates based on just domestic R&D) to .24 percent Since information on the country of origin of citations is available only for 1976 and subsequent years, the productivity calculations mentioned above estimate the foreign percent of citations for years prior to 1976. Since information on foreign citations is more reliable for 1973-2002, we mention that, in these years, the domestic R&D contribution of .18 percent a year to productivity growth increased to .29 percent a year when foreign R&D is taken into account. Foreign R&D is important, of course, due to the rapid growth of foreign citations, as reflected in Table 2.

⁶⁷ Hall concludes (1993, page 317) that greater competition, much of it foreign, reduced the U.S. private return to R&D in the 1980's, but had less of an impact on social returns to R&D. As R&D increased in other countries, the United States perhaps lost some monopoly power over innovation (Freeman (2005)). ⁶⁸ In developing countries, technology typically adopts and modifies advances created in the wealthy nations, so information on innovations, such as patents, provides a less complete picture of new technology.

On the basis of the evidence considered, privately financed R&D in industry should be treated as an investment and included in the relevant R&D stock. Returns to R&D are very high, but these high returns accrue <u>only</u> to privately financed R&D. Many elements of university and government research have very low returns, overwhelmingly contribute to economic growth only indirectly, if at all, and do not belong in investment.

Much remains to be understood about the specific pathways through which R&D in advanced nations is transmitted to developing countries. There are many potential pathways through which this knowledge may spread. Once R&D is included in the accounts, economists will hopefully pay much more attention to understanding exactly which channels are effective in transmitting knowledge to poorer countries.

REFERENCES

Acharya, Ram C. and Keller, Woflgang (2007), "Technology Transfer Through Imports," National Bureau of Economic Research Working Paper 13086, May 2007.

Adams, James D. (2004), "Learning, Internal Research and Spillovers: Evidence from a Sample of R&D Laboratories," Working Paper 0409, Rensselaer Polytechnic Institute, March 2004.

Adams, James D., Chiang, Eric P. and Jensen, Jeffrey L. (2003), "The Influence of Federal Laboratory R&D on Industrial Research," <u>Review of Economics and Statistics</u> (November 2003), pp. 1003-1020.

Aitken, Brian J. and Harrison, Ann E. (1999), "Do Domestic Firms Benefit from Foreign Direct Investment? Evidence from Venezuela," <u>American Economic Review</u> (June 1999), pp. 605-618.

Almeida, Paul (2004), "Using Patent Data to Track the Flow of Technological Knowledge and the Mobility of Scientists," paper presented to National Economists Club, Washington, D.C, August 2004.

AlAzzawi, Shireen (2004), "Foreign Direct Investment and Knowledge Flows: Evidence from Patent Citations," University of California, Davis, January 2004.

Australian Bureau of Statistics (2004), "Capitalising research and development", paper presented at meeting of Canberra II Group, Washington, D.C., March 17-19, 2004.

Barro, Robert (1998), "Notes on Growth Accounting," National Bureau of Economic Research Working Paper 6654, July 1998.

Bartel, Ann P. and Harrison, Ann E. (2005), "Ownership Versus Environment: Disentangling the Sources of Public Sector Inefficiency," <u>Review of Economics and</u> <u>Statistics</u> (February 2005), pp. 135-147.

Basant, Rakesh and Fikkert, Brian (1996), "The Effects of R&D, Foreign Technology Purchase, and Domestic and International Spillovers on Productivity in Indian Firm," <u>Review of Economics and Statistics</u> (May 1996), pp. 187-199.

Baumol, William F. (2000), <u>The Free Market Innovation Machine: Analyzing the Growth</u> <u>Miracle of Capitalism</u>, Princeton University Press, Princeton, 2000.

Bernstein, Jeffrey and Nadiri, M. Ishaq (1988), "Interindustry Spillovers, Rates of Return, and Production in High-Tech Industries," <u>American Economic Review</u> (May 1988), pp. 429-434.

Bernstein, Jeffrey and Nadiri, M. Ishaq (1991), "Product Demand, Cost of Production, Spillovers, and the Social Rate of Return to R&D," National Bureau of Economic Research Working Paper 3625, Cambridge, 1991.

Blalock, Garrick and Veloso, Francisco (2006), "Imports, Productivity, and Supply Chain Learning," Cornell University, September 2006.

Bloom, Nick, Sudun, Raffaela, and van Reenen, John (2006), "IT Ain't What You Do, IT's the Way You Do I.T.: Investigating the Productivity Miracle Using the Overseas Activities of U.S. Multinationals," paper presented at the Center for Economic Studies, March 2006.

Blundell, Richard, Griffith, Rachel, and Van Reenen, John (1999), "Market Share, Market Value and Innovation in a Panel of British Manufacturing Firms," <u>Review of</u> <u>Economic Studies</u> (July 1999), pp. 529-554.

Bonte, Werner (2003), "Does Federally Financed Business R&D Matter for U.S. Productivity Growth?" <u>Applied Economics</u> (October 2003), pp. 1619-1625.

Borensztein, Eduardo, De Gregorio, Jose and Lee, Jong-Wha (1998), "How Does Foreign Direct Investment Affect Economic Growth?" Journal of International Economics (June 1998), pp. 115-135.

Branstetter, Lee (2000), "Is Foreign Direct Investment a Channel of Knowledge Spillovers? Evidence from Japan's FDI in the United States," National Bureau of Economic Research Working Paper 8015, Cambridge, November 2000.

Bresnahan, Timothy (1999), "Computerization and Wage Dispersion: An Analytical Reinterpretation", <u>Economic Journal</u> (June 1999), pp. F390-F415.

Bresnahan, Timothy and Greenstein, Shane (1996), "Technical Progress and Co-Invention in Computing and in the Uses of Computers," <u>Brooking Papers on Economic</u> <u>Activity, Microeconomics</u> (1996), pp. 1-83.

Brown, Lawrence D., Plewes, Thomas J., and Gerstein, Marisa A. (2004), <u>Measuring</u> <u>Research and Development Expenditures in the U.S. Economy</u>, National Research Council, Washington, D.C., 2004.

Brynjolfsson, Eric, and Hitt, Loren, (2003), "Computing Productivity: Firm-Level Evidence", <u>Review of Economics and Statistics</u> (November 2003), pp. 793-808.

Brynjolfsson, Eric, Hitt, Loren, and Yang, Shinkyu (2002), "Intangible Assets: Computers and Organizational Capital", <u>Brookings Papers on Economic Activity</u> (Macroeconomics 1, 2002), pp. 137-199.

Caballero, Ricardo and Jaffe, Adam (1993), "How High are the Giant's Shoulders: An Empirical Assessment of Knowledge Spillovers and Creative Destruction in a Model of Economic Growth," <u>NBER Macroeconomics Annual</u>, 1993, pp. 15-73.

Canberra Group (2003) on the Measurement of Non-Financial Assets, Second Meeting, 13-15 October 2003, Paris. Items 91 to 102 clearly state that "R&D should indeed be capitalized in National Accounts".

Canberra II Group (2006), "Capitalization of R&D," Joint Expert Meeting, March 24, 2006.

Carson, Carol S., Grimm, Bruce T. and Moylan, Carol E. (1994), "A Satellite Account for Research and Development", <u>Survey of Current Business</u> (November 1994), pp. 37-71.

Caselli, Francesco and Wilson, Daniel (2004), "Importing Technology," <u>Journal of</u> <u>Monetary Economics</u> (January 2004), pp. 1-32.

Coe, David T. and Helpman, Elhanan (1995), "International R&D Spillovers", <u>European</u> <u>Economic Review</u>, (May 1995), pp. 859-887.

Coe, David T., Helpman, Elhanan and Hoffmaister, Alexander (1997), "North-South R&D Spillovers", <u>Economic Journal</u> (January 1997), pp. 134-149.

Connolly, Michelle (2003), "The Dual Nature of Trade: Measuring Its Impact on Imitation and Growth," Journal of Development Economics (October 2003), pp. 31-55.

Corrado, Carol, Hulten, Charles, and Sichel, Daniel (2003), "Measuring Capital and Technology: An Expanded Framework," presented at Conference on Research and Wealth, Washington, D.C.

Cuneo, Philippe and Mairesse, Jacques (1984), "Productivity and R&D at the Firm Level in French Manufacturing," pp. 375-392 in Griliches, Zvi (Ed.), <u>R&D, Patents, and</u> <u>Productivity</u>, University of Chicago Press, Chicago, 1984.

Darby, Michael R. and Zucker, Lynne G. (2003), "Growing by Leaps and Inches: Creative Destruction, Real Cost Reduction and Inching Up," <u>Economic Inquiry</u> (January 2003), pp. 1-19.

de Haan, Mark (2004), "Progress Report on R&D capitalization: The Netherlands", paper presented at meeting of Canberra II Group, Washington, D.C., March 17-19, 2004.

Djankov, Simeon, La Porta, Rafael, Lopez-de-Silanes, Florencio, and Shleifer, Andrei (2003), "Courts," <u>Quarterly Journal of Economics</u> (May 2003), pp. 453-517.

Eaton, Jonathan and Kortum, Samuel (1996), "Trade in Ideas: Patenting and Productivity in the OECD," Journal of International Economics, (May 1996), pp. 251-278.

Eaton, Jonathan and Kortum, Samuel (1999), "International Technology Diffusion: Theory and Measurement," <u>International Economic Review</u> (August 1999), pp. 537-570.

Frantzen, Dirk (2002), "Intersectoral and International R&D Knowledge Spillovers and Total Factor Productivity," <u>Scottish Journal of Political Economy</u> (No. 3, 2002), pp. 531-545.

Fraumeni, Barbara and Okubo, Sumiye (2005), "R&D in the National Income and Product Accounts: A First Look at its Effect on GNP", pp. 275-316 in Corrado, Carol, Haltiwanger, John, and Sichel, Daniel (Eds.), <u>Measuring Capital in the New Economy</u>, National Bureau of Economic Research, University of Chicago Press, Chicago, 2005.

Freeman, Richard B. (2005), "Does Globalization of the Scientific/Engineering Workforce Threaten U.S. Economic Leadership?", National Bureau of Economic Research Working Paper 11457, July 2005.

Fung, Michael K. (2004), "Technological Opportunity and Productivity of R&D Activities," Journal of Productivity Analysis (March 2004), pp.167-181.

Funk, Mark (2001), "Trade and International R&D Spillovers among OECD Countries," <u>Southern Economic Journal</u> (January 2001), pp. 725-736.

Gordon, Robert J. (1990), <u>The Measurement of Durables Goods Prices</u>, National Bureau of Economic Research, University of Chicago Press, Chicago, 1990.

Gorg, Holger and Strobl, Eric (2001), "Multinational Companies and Productivity Spillovers: A Meta-Analysis," <u>Economic Journal</u> (November 2001), pp. F723-F739.

Goto, Akira and Suzuki, Kazuyuki (1989), "R&D Capital, Rate of Return on R&D Investment, and Spillover of R&D in Japanese Manufacturing Industries," <u>Review of</u> <u>Economics and Statistics</u> (November 1989), pp. 555-564.

Grabowski, Henry G. and Mueller, Dennis C. (1978), "Industrial Research and Development, Intangible Capital Stocks, and Firm Profit Rates," <u>Bell Journal of Economics</u> (Autumn 1978), pp. 328-343.

Greenwood, Jeremy, Hercowitz, Zvi, and Krusell, Per (1997), "Long-Run Implications of Investment-Specific Technological Change," <u>American Economic Review</u> (June 1997), pp. 342-362.

Griffith, Rachel, Redding, Stephen, and Van Reenen, John (2004), "R&D and Absorptive Capacity: From Theory to Data," <u>Scandinavian Journal of Economics</u>, forthcoming, 2004).

Griliches, Zvi (1979), "Issues in Assessing the Contribution of Research and Development to Productivity Growth," <u>Bell Journal of Economics</u>, Spring 1979, pp. 92-116.

Griliches, Zvi (1980), "Returns to Research and Development Expenditures in the Private Sector," pp. 419-454 in Kendrick, J.W. and Vaccara, B. (Eds.), <u>New Developments in</u> <u>Productivity Measurement</u>, National Bureau of Economic Research, University of Chicago Press, Chicago, 1980.

Griliches, Zvi (1986), "Productivity, R&D, and Basic Research at the Firm Level in the 1970's," <u>American Economic Review</u> (March 1986), pp. 141-154

Griliches, Zvi (1990), "Patent Statistics as Economic Indicators: A Survey," Journal of Economic Literature (December 1990), pp. 1661-1707.

Griliches, Zvi (1992), "The Search for R&D Spillovers," <u>Scandinavian Journal of</u> <u>Economics</u> (Supplement 1992), pp. S29-S47.

Griliches, Zvi (2000), <u>R&D</u>, <u>Education</u>, and <u>Productivity: A Retrospective</u>, Harvard University Press, Cambridge, 2000.

Griliches, Zvi and Lichtenberg, Frank (1984a), "Interindustry Technology Flows and Productivity Growth: A Reexamination", <u>Review of Economics and Statistics</u> (May 1984), pp. 324-329.

Griliches, Zvi and Lichtenberg, Frank (1984b), "R&D and Productivity Growth at the Industry Level: Is There Still a Relationship", pp. 465-496 in Griliches, Zvi (Ed.), <u>R&D</u>, <u>Patents, and Productivity</u>, National Bureau of Economic Research, University of Chicago Press, Chicago, 1984.

Griliches, Zvi and Mairesse, Jacques (1984), "Productivity and R&D at the Firm Level,", pp. 339-374 in Griliches, Zvi (Ed.), <u>R&D</u>, <u>Patents</u>, and <u>Productivity</u>, National Bureau of Economic Research, University of Chicago Press, Chicago, 1984.

Guellec, Domininque and van Pottelsberge de la Potterie, Bruno (2004), "From R&D to Productivity Growth: Do the Institutional Settings and the Source of Funds of R&D Matter?" <u>Oxford Bulletin of Economics and Statistics</u> (July 2004), pp. 353-378.

Hall, Bronwyn H. (1993), "Industrial Research during the 1980s: Did the Rate of Return Fall?", <u>Brookings Papers on Economic Activity, Microeconomics</u>, 1993, Number 2, pp. 289-343.

Hall, Bronwyn H., Jaffe, Adam B, and Trajtenberg, Manuel (2005), "Market Value and Patent Citations," <u>Rand Journal of Economics</u>, 2005, pp. 16-38.

Hall, Robert E. and Jorgenson, Dale W. (1967), "Tax Policy and Investment Behavior," <u>American Economic Review</u> (June 1967), pp. 391-414.

Hallak, Juan Carlos and Levinsohn, James, "Fooling Ourselves: Evaluating the Globalization and Growth Debate," NBER Working Paper 10244, Cambridge, January 2004.

Harhoff, Dietmar, Narin, Francis, Scherer, Frederic M. and Vopel, Katrin (1999), "Citation Frequency and the Value of Patented Inventions," <u>Review of Economics and</u> <u>Statistics</u> (August 1999), pp. 511-515.

Haskel, Jonathan E., Pereira, Sonia C. and Slaughter, Matthew J. (2007), "Does Inward Foreign Direct Investment Boost the Productivity of Domestic Firms," <u>Review of</u> <u>Economics and Statistics</u> (August 2007), pp. 482-496.

Jaffe, Adam B. (1986), "Technological Opportunity and Spillovers of R&D: Evidence from Firm's Patents, Profits, and Market Value", <u>American Economic Review</u> (December 1986), pp. 984-1001.

Jaffe, Adam B. (1988), "Demand and Supply Influences in R&D Intensity and Productivity Growth", <u>Review of Economics and Statistics</u> (August 1988), pp. 431-437.

Jaffe, Adam B., Fogarty, Michael S. and Banks, Bruce A. (1998), "Evidence from Patents and Patent Citations on the Impact of NASA and Other Federal Labs on Commercial Innovation," Journal of Industrial Economics (June 1998), pp. 183-205.

Jaffe, Adam and Trajtenberg, Manuel (2002), <u>Patents, Citations and Innovations</u>, MIT Press, Cambridge, 2002.

Jorgenson, Dale W. (1966), "The Embodiment Hypothesis," Journal of Political Economy (February 1966), pp. 1-17.

Jorgenson, Dale W. and Griliches, Zvi (1967), "The Explanation of Productivity Change," <u>Review of Economic Studies</u> (July 1967), pp. 249-283.

Kao, Chihwa, Chiang, Min-Hsien and Chen, Bangtian (1999), "InternationalR&D Spillovers: An Application of Estimation and Inference in Panel Cointegration," <u>Oxford</u> <u>Bulletin of Economics and Statistics</u> (November 1999), pp. 693-711.

Keller, Wolfgang (2002), "Geographic Localization of International Technological Diffusion", <u>American Economic Review</u> (March 2002), pp. 120-142.

Keller, Wolfgang (2004), "International Technology Diffusion," <u>Journal of Economic</u> <u>Literature</u> (September 2004), pp. 752-782.

King, Robert J. and Levine, Ross (1993), 'Finance and Growth: Schumpeter Might be Right," <u>Quarterly Journal of Economics</u> (August 1993), pp. 717-737.

Klevorick, Alvin K., Levin, Richard C., Nelson, Richard R., and Winter, Sidney G. (1995), "On the Sources and Significance of Interindustry Differences in Technological Opportunities" <u>Research Policy</u> (March 1995), pp. 185-205.

Kortum, Samuel S. (1997), "Research, Patenting, and Technological Change," <u>Econometrica</u> (November 1997), pp. 1389-1414.

Lanjouw, Jean O. and Schankerman, Mark (2004), "Patent Quality and Research Productivity: Measuring Innovation with Multiple Indicators," <u>Economic Journal</u> (April 2004), pp. 441-465.

La Porta, Rafael, Lopez-de-Silanes, Florencio, Shleifer, Andrei, and Vishny, Robert (1998), "Law and Finance," <u>Journal of Political Economy</u> (December 1998), pp. 1113-1155.

Lehr, William and Lichtenberg, Frank R. (1998), "Computer Use and Productivity Growth in Federal Government Agencies, 1987-92," <u>Journal of Industrial Economics</u> (June 1998), pp. 257-279.

Leonard, William N. (1971), "Research and Development in Industrial Growth," Journal of Political Economy (March 1971), pp. 232-256

Lewbel, Arthur (1997), "Constructing Instruments for Regressions with Measurement Error when No Additional Data Are Available, With An Application to Patents and R&D," <u>Econometrica</u> (September 1997), pp. 1201-1213.

Lewis, William L. (2004), <u>The Power of Productivity</u>, University of Chicago Press, Chicago, 2004.

Lichtenberg, Frank R. and Siegel, Donald (1991), "The Impact of R&D Investment on Productivity: New Evidence Using Linked R&D-LRD Data," <u>Economic Inquiry</u> (April 1991), pp. 203-229.

Lichtenberg, Frank R. and van Pottelsberge de la Potterie, Bruno (1998), "International R&D Spillovers: A Comment", <u>European Economic Review</u> (September 1998), pp. 1483-1491.

Lichtenburg, Frank R. and Virabhak, Suchin (2002), "Pharmaceutical-Embodied Technical Progress, Longevity and Quality of Life", NBER Working Paper 9351, November 2002.

Luintel, Kul B. and Khan, Mohasid (2004), "Are International R&D Spillovers Costly for the U.S.?" <u>Review of Economics and Statistics</u> (November 2004), pp. 896-910.

Mansfield, Edwin (1985), "How Rapidly Does New Industrial Technology Leak Out?" Journal of Industrial Economics (December 1985), pp. 217-223.

Mansfield, Edwin (1995), "Academic Research Underlying Industrial Innovations: Sources, Characteristics, and Financing," <u>Review of Economics and Statistics</u> (February 1995), pp. 55-65.

Mansfield, Edwin, Rapoport, John, Romeo, Anthony, Wagner, Samuel, and Beardsley, George (1977) "Social and Private Rates of Return from Industrial Innovations", <u>Quarterly Journal of Economics</u> (May 1977), pp. 221-240.

Mansfield, Edwin, Rapoport, John, Schnee, Jerome, Wagner, Samuel, and Hamburger, Michael (1971), <u>Research and Innovation in the Modern Corporation</u>, W.W. Norton & Company, New York, 1971.

Maursith, P.B. and Verspagen, B. (2002), "Knowledge Spillovers in Europe: A Patent Citations Analysis," <u>Scandinavian Journal of Economics</u> (December 2002), pp. 531-545.

Mohnen, P. (1996), "R&D Externalities and Productivity Growth," <u>STI Review</u>, vol. 18, pp. 39-66.

Moretti, Enrico (2004), "Workers Education, Spillovers, and Productivity: Evidence from Plant-Level Production Functions," <u>American Economic Review</u> (June 2004), pp. 656-690.

Muendler, Marc-Andreas (2004), "Trade, Technology, and Productivity: A Study of Brazilian Manufacturers, 1986-1998," CESIfo Working Paper No. 1148, March 2004.

Nadiri, M. Ishaq (1993), "Innovations and Technological Spillovers," National Bureau of Economic Research Working Paper 4423, August 1993.

Nathan, Robert R.Associates (1978), <u>Net Rates of Return on Innovations</u>, Report to the National Science Foundation, Arlington, Virginia, October 1978.

Nelson, Richard R. (1988) "Modeling the Connections in the Cross Section Between Technical Progress and R&D Intensity", <u>Rand Journal of Economics</u> (August 1988), pp. 478-485.

Nelson, Richard R. and Pack, Howard (1999), "The Asian Miracle and Modern Growth Theory," <u>Economic Journal</u> (July 1999), pp. 416-436.

Nordhaus, William D. (2003), "The Health of Nations: The Contribution of Improved Health to Living Standards," pp. 41-73 in Murphy, Kevin M. and Topel, Robert H. (Eds.), <u>Measuring the Gains from Medical Research: An Economic Approach</u>, University of Chicago Press, Chicago, 2003.

Nordhaus, William D. (2004), "Schumpeterian Profits in the American Economy: Theory and Measurement," National Bureau of Economic Research Working Paper 10433, Cambridge, April 2004.

OECD (2003), The Sources of Economic Growth in OECD Countries, Paris 2003.

Okubo, Sumiye, Robbins, Carol A., Moylan, Carol E., Sliker, Brian K., Schultz, Laura I. and Mataloni, Lisa S. (2006a), "R&D Satellite Account: Preliminary Estimates," Bureau of Economic Analysis September 28, 2006.

Okubo, Sumiye, Robbins, Carol A., Moylan, Carol E., Sliker, Brian K., Schultz, Laura I. and Mataloni, Lisa S. (2006b), "BEA's Research and Development Satellite Account," <u>Survey of Current Business</u> (December 2006), pp. 14-46.

Olley, G. Steven and Pakes, Ariel (1996), "The Dynamics of Productivity in the Telecommunication Equipment Industry," <u>Econometrica</u> (November 1996), pp. 1263-1297.

Parente, Stephen L. and Prescott, Edward C. (1994), "Barriers to Technology Adoption and Development," Journal of Political Economy (April 1994), pp. 298-321.

Parente, Stephen L. and Prescott, Edward C. (1999), "Monopoly Rights: A Barrier to Riches," <u>American Economic Review</u> (December 1999), pp. 1216-1233.

Park, Jungsoo (2004), "International and Intersectoral R&D Spillovers in the OECD and East Asian Economies," <u>Economic Inquiry</u> (October 2004), pp. 739-757.

Peri, Giovanni (2005), "Determinants of Knowledge Flows and Their Effect on Innovation," <u>Review of Economics and Statistics</u> (May 2005), pp.308-322.

Piekarz, Rolf (1983), "R&D and Productivity Growth: Policy Studies and Issues," <u>American Economic Review</u> (May 1983), pp. 210-214.

Robbins, Carol A. (2006), "R&D Expenditures for the U.S.: A Frascati to System of National Accounts Application to U.S. Data," February 28, 2006.

Romer, Paul (1990), "Endogenous Technical Change," <u>Journal of Political Economy</u> (October 1990, pp. S71-S102.

Savvides, Andreas and Zachariadis, Marios (2004), "International Technology Diffusion and the Growth of TFP in the Manufacturing Sector of Developing Economies," April 2004, <u>Review of Development Economics</u>, forthcoming.

Schankerman, Mark (1981), The Effect of Double-Counting and Expensing on the Measured Returns to R&D," <u>Review of Economics and Statistics</u> (August 1981), pp. 454-458.

Scherer, Frederic M. (1982), "Inter-Industry Technology Flows and Productivity Growth," <u>Review of Economics and Statistics</u> (November 1982), pp. 627-634.

Scherer, Frederic M. (1984), "Using Linked Patent and R&D Data to Measure Interindustry Technology Flows," pp. 417-464 in Griliches, Zvi (Ed.), <u>R&D</u>, <u>Patents</u>, and <u>Productivity</u>, University of Chicago Press, Chicago, 1984.

Scherer, Frederic M. (1999), <u>New Perspectives on Economic Growth and Technological</u> <u>Innovation</u>, Brookings Institution Press, Washington, D.C., 1999.

Scherer, Frederic M. (2003), "Technology Flows Matrix Estimation Revisited," <u>Economic Systems Research</u> (September 2003), pp. 327-358.

Schiff, Maurice and Wang, Yanling (2004), "On the Quantity and Quality of Knowledge: The Impact of Openness and Foreign Research and Development on North-North and North-South Technology Spillovers," World Bank Policy Research Working Paper 3190, January 2004. Schiff, Maurice, Wang, Yanling, and Olarreaga, Marcelo (2002), "Trade Related Technology Diffusion and the Dynamics of North-South and South-South Integration," World Bank Policy Research Working Paper 2861, June 2002.

Schultz, Laura I. (2006), "Estimating Domestic Returns and Spillovers from Business R&D," paper delivered at Western Economics Assocation meetings, June 2006.

Scully, Gerald W. (1988), "The Institutional Framework and Economic Development," Journal of Political Economy (June 1988), pp. 652-662.

Sena, Vania (2004), "The Return of the Prince of Denmark: A Survey of Recent Developments in the Economics of Innovation," <u>Economic Journal</u> (June 2004), pp F312-F332.

Sveikauskas, Leo (1981), "Technology Inputs and Multifactor Productivity Growth," <u>Review of Economics and Statistics</u> (May 1981), pp. 275-282.

System of National Accounts, 1993, prepared by the Inter-Secretariat Working Group on National Accounts, 1993.

Terleckyj, Nestor (1974), <u>Effects of R&D on the Productivity Growth of Industries: An</u> <u>Exploratory Study</u>, National Planning Association, Washington, 1974.

Tewksbury, J.G., Crandall, M.S., and Crane, W.E., (1980), "Measuring the Societal Benefits of Innovations", <u>Science</u> (August 8, 1980), pp. 658-662.

Toole, Andrew (2007), "Does Public Biomedical Research Complement Private Pharmaceutical Research and Development Investment?" <u>Journal of Law and Economics</u> (February 2007), pp. 81-104.

Trajtenberg, Manuel (1990), "A Penny for Your Quotes: Patent Citations and the Value of Innovations," <u>RAND Journal of Economics</u> (Spring 1990), pp. 172-187.

U.S. Bureau of Labor Statistics, <u>The Impact of Research and Development on</u> <u>Productivity Growth</u>, Bulletin 2331, Government Printing Office, Washington, D.C., September 1989.

Verspagen, Bart (1997a), "Measuring Intersectoral Technology Spillovers: Estimates from the European and US Patent Office Databases," <u>Economic Systems Research</u> (1997), pp. 47-65.

Verspagen Bart (1997b), "Estimating International Technology Spillovers Using Technology Flow Matrices," <u>Weltwirtschaftliches Archiv</u> (1997, Number 2), pp. 226-248.

Wolfe, Raymond M. (2003), "U.S. Industry Sustains R&D Expenditures During 2003 Despite Decline in Performers' Aggregate Sales," NSF04-301, October 2003.

Wolff, Edward N. (1997), "Spillovers, Linkages, and Technical Change," <u>Economic</u> <u>Systems Research</u> (March 1997), pp. 9-23.

Xu, Bin and Wang, Jianmao (1999), "Capital Goods Trade and R&D Spillovers in the OECD," <u>Canadian Journal of Economics</u> (November 1999), pp. 1258-1274.

Zucker, Lynne G., Darby, Michael R. and Armstrong, Jeff (1998) "Geographically Localized Knowledge: Spillovers or Markets?" <u>Economic Inquiry</u> (January 1998), pp. 65-86.